Author's response to reviews

Title: Spontaneous rupture of an infected renal cyst and external drainage through lumbar surgical scar in a male patient with cervical spinal cord injury: a case report

Authors:

Subramanian Vaidyanathan (S.Vaidyanathan@southportandormskirk.nhs.uk)
Peter L Hughes (Peter.Hughes@southportandormskirk.nhs.uk)
Tun Oo (Tun.Oo@southportandormskirk.nhs.uk)
Bakul M Soni (bakul.Soni@southportandormskirk.nhs.uk)

Version: 3 Date: 24 May 2007

Author’s response to reviews:

Claire Shaw

Assistant Editor - Independent Journals

Dear Ms Shaw

I wish to thank you for your e-mail.

We amended the manuscript as per your advice.

Title was revised. Acknowledgement section was included. Authors have declared that they do not have any competing interests. The contribution made by each author is stated in the manuscript. Two figures have been deleted so that the revised manuscript contains only three illustrations.

Thank you for your help

I am most grateful to you for considering this manuscript for publication.

Yours sincerely

S Vaidyanathan